Author's response to reviews

Title: Socio-economic patterns of overweight, obesity but not thinness persist from childhood to adolescence in a 6-year longitudinal cohort of Australian schoolchildren from 2007 to 2012

Authors:

Jennifer A O'Dea (jennifer.odea@sydney.edu.au)
Hueiwen Chiang (hchi1507@uni.sydney.edu.au)
Louisa Peralta (louisa.peralta@sydney.edu.au)

Version: 6 Date: 19 February 2014

Author's response to reviews:

The revisions requested by the Editors have been added - Competing interests and Contribution by authors.

There were no other requests from the Reviewers.